Math 230, Spring 1999 Mathematica assignment 2 (due Monday, 2/8/99)

The following problems are taken from pages 59 to 67 of the Differential Equations with Mathematica book. Please put some effort into presenting your work well. That is, include comments where appropriate, and use the Format menus to distinguish text from headings from actual computation. It might work best just to do all the work and then go back and insert comments, format things right, etc. Also, remember that when the instructions don’t explicitly require the use of Mathematica, you’re free to use pencil and paper to do some of the work--whatever gets the job done most effectively.

ˇ Problem 2.

ˇ Problem 4.

ˇ Problem 10.

ˇ Problem 12.

ˇ Problem 16.

ˇ Problem 17.

assign2.nb 1